Author’s response to reviews

Title: Fibromyalgia in the workplace: risk factors for sick leave are related to professional context rather than fibromyalgia characteristics. A French national survey of 955 patients

Authors:
Francoise Laroche (francoise.laroche2@aphp.fr)
Deborah Azoulay (deborahazoulay@hotmail.fr)
Anne Priscille Trouvin (aptrouvin@gmail.com)
Joel Coste (joel.coste@aphp.fr)
Serge Perrot (serge.perrot@aphp.fr)

Version: 4 Date: 05 Aug 2019

Author’s response to reviews:

Dear Editor,

Please find enclosed the revised manuscript: «Fibromyalgia in the workplace: risk factors for sick leave are related to professional context rather than fibromyalgia characteristics. A French national survey of 955 patients” - BRHM - D-19-00020R1

We have amended our manuscript according to your requests. We would be grateful if you would consider publishing this paper in its revised form.

Yours sincerely,

Prof. Françoise Laroche